\begin{table}%t1
\caption{\label{cflib} Stellar atmospheric parameters for six CFLIB stars of different spectral types.}
%%\centerline

 {
\begin{tabular}{lccccc}
\hline\hline
Name    & Sp. type & \teff{} & \logg{} & [Fe/H] & Ref.     \\
%\multicolumn{1}{c}{(1)} & (2) & (3) & (4) & (5) & (6)          \\
	&	  & (K)	    &  g~cm$^{-2}$~s$^{-1}$  & (dex)  &	(1)	\\
\hline
   HD 30614	& O9.5Iae	& 29647	        & 3.05	        &  0.30      & 1  \\
		&		& 33972  	& 3.18		& --0.05	     &  	\\
  HD 195324      & A1Ib 	        & 9300	        & 1.90		& --0.11      & 2  \\
    		&		& 9847		& 1.94		& --0.16	     &  \\
  HD 114642      & F5.5V 	& 6434	        & 3.83	        & --0.12      & 3  \\
    		&		& 6431		& 4.04		& --0.12	     &  	\\
   HD 76151 	& G2V 	        & 5768	        & 4.45	        &  0.06      & 4  \\
     		&		& 5728		& 4.41		&  0.09	     &          \\
   HD 10780	& K0V 	        & 5359	        & 4.44	        &  0.02      & 4  \\
     		&		& 5330		& 4.50		&  0.06	     &	        \\
   HD 42475 	& M1Iab 	& 4000	        & 0.70		& --0.36      & 5  \\
     		&		& 3988		& 0.32		&  0.02	     &        	\\
\hline
\end{tabular}}
\medskip
\par
The atmospheric parameters on the first line are compiled from the literature, and
on the second line fitted by \ulyss.\\
(1) Sources for the atmospheric parameters:
[1] \citet{takada77}; [2] \citet{venn95}; [3]
\citet{takeda07}; [4] \citet{soubiran08}; [5] \citet{luck80}.
\end{table}